Author's response to reviews

Title: The prevalence of clinically-relevant comorbid conditions in patients with COPD: a cross-sectional study using data from NHANES 1999-2008

Authors:

Kerry M Schnell (kschnel1@jhmi.edu)
Carlos O Weiss (cweiss9@jhmi.edu)
Todd Lee (toddlee@uic.edu)
Jerry A Krishnan (jakris@uic.edu)
Bruce Leff (bleff@jhmi.edu)
Jennifer L Wolff (jwolff@jhsphs.edu)
Cynthia Boyd (cyboyd@jhmi.edu)

Version: 2 Date: 13 November 2011

Author's response to reviews:

Previously, we submitted this manuscript to BMC Medicine. The editors suggested that our manuscript may be better suited for BMC Pulmonary Medicine. Revisions were not suggested and revisions have not been made for the resubmission of this manuscript.